Author’s response to reviews

Title: Doctoral Level Research and Training Capacity in the Social Determinants of Health at universities and higher education institutions in India, China, Oman and Vietnam: A survey of needs

Authors:

Farhad Ali (farhadali1@gmail.com)

Arun Shet (arunshet@sjri.res.in)

Weirong Yan (weirongy@gmail.com)

Abdullah Al-Maniri (abdullah.almaniri@trc.gov.om)

Salla Atkins (salla.atkins@ki.se)

Henry Lucas (h.lucas@ids.ac.uk)

Version: 1 Date: 01 Mar 2017

Author’s response to reviews:

Dear Editor,

The paper has been thoroughly edited based on the comments of the reviewers. All the desired changes have been incorporated in the revised manuscript. Kindly acknowledge the receipt of submission of the revised manuscript.

best regards

Farhad Ali

Lead Author